Author’s response to reviews

Title: Reference data on hand grip and lower limb strength using the Nintendo Wii Balance Board: A cross-sectional study of 354 subjects from 20 to 99 years of age

Authors:

Fredrik Eika (eikafredrik@gmail.com)
Andreas Blomkvist (andreas.wahl@gmail.com)
Martin Rahbek (martin.torp.rahbek@rsyd.dk)
Karin Eikhof (karin.dam@rn.dk)
Mette Hansen (metteduush@gmail.com)
Malene Søndergaard (malene-soendergaard@hotmail.com)
Jesper Ryg (Jesper.Ryg@rsyd.dk)
Stig Andersen (stiga@dadalnet.dk)
Martin Jørgensen (mgj@rn.dk)

Version: 2 Date: 03 Jan 2019

Author’s response to reviews:

As requested by the editor, this is the manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Figures are uploaded as separate files.